Author's response to reviews

Title: Access to electronic health knowledge in five countries in Africa: a descriptive study

Authors:

Helen Smith (cjdhel@liv.ac.uk)
Hasifa Bukirwa (hbukirwa@liverpool.ac.uk)
Oscar Mukasa (oscarmukasa@hotmail.com)
Paul Snell (psnell@mrc.gm)
Sylvester Adeh-Nsoh (adeh_nsoh@yahoo.com)
Selemani Mbuyita (mbuita@yahoo.com)
Masanja Honorati (honorati.masanja@unibas.ch)
Bright Orji (brightorji@hotmail.com)
Pual Garner (pgarner@liv.ac.uk)

Version: 3 Date: 3 May 2007

Author's response to reviews:

1. We have removed capital letters for country names.
2. We have removed the running title, key words and word count.
3. We have removed the 'objectives' heading from the abstract (and added the sentence to the background)
4. We have changed the heading 'introduction' to 'background'.
5. We have amended 'authors' contributions'.
6. We have put the additional files section at the end of the document and re-formatted these.
7. We have added a heading 'tables' to the section containing tables.